Author's response to reviews

Title: Diagnostic accuracy of basal TSH determinations based on the intravenous TRH stimulation test. A retrospective evaluation of 2570 tests and comparison with the literature.

Authors:

Helga Moncayo (anmeldung@womed.at)
Otto Dapunt (anmeldung@womed.at)
Roy Moncayo (anmeldung@womed.at)

Version: 2 Date: 20 August 2006

Author's response to reviews:

The length of the abstract was reduced to 285 words.